Author’s response to reviews

Title: Virtual House Calls for Parkinson disease (Connect.Parkinson): study protocol for a randomized, controlled trial

Authors:

Meredith A Achey (meredith.achey@chet.rochester.edu)
Christopher A Beck (beck@bst.rochester.edu)
Denise B Beran (dberan@parkinson.org)
Cynthia M Boyd (cyboyd@jhmi.edu)
Peter N Schmidt (pschmidt@parkinson.org)
Allison W Willis (Allison.Willis@uphs.upenn.edu)
Sara S Riggare (sara@riggare.se)
Richard B Simone (rbsimone@pacbell.net)
Kevin M Biglan (kevin_biglan@urmc.rochester.edu)
E. Ray Dorsey (ray.dorsey@chet.rochester.edu)

Version: 2 Date: 6 August 2014

Author's response to reviews:

A summary of the revisions has been included as a comment to the editors.